Author's response to reviews

Title: First Look: A Cluster-randomized trial of ultrasound to improve pregnancy outcomes in low income country settings

Authors:

Elizabeth M McClure (mcclure@rti.org)
Robert O Nathan (ronathan@uw.edu)
Sarah Saleem (sarah.saleem@aku.edu)
Hillary Mabeya (hmabeya@yahoo.com)
Ana Garces (ana.garces@imsalud.edu)
Elwyn Chomba (echomba@zamnet.zm)
Antoinette Tshefu (antotshe@yahoo.com)
David Swanson (dadley@yahoo.com)
Fabian Esamai (fesamai2007@gmail.com)
Lester Figuero (lesterfigueroa@imsalud.org)
Waseem Mirza (waseem.mirza@aku.edu)
David Muyodi (dmuyodi6@gmail.com)
Holly Franklin (hfranklin@rti.org)
Adrien Lokangaka (alokangaka@gmail.com)
Dieudonne Bidashimwa (bidashimwa@gmail.com)
Omrana Pasha (omrana.pasha@aku.edu)
Musaku Mwenechanya (mmusaku@gmail.com)
Carl L Bose (cbose@unc.edu)
Waldemar A Carlo (wcarlo@peds.uab.edu)
Michael Hambidge (Michael.hambidge@ucdenver.edu)
Edward A Liechty (eliecht@iupui.edu)
Nancy Krebs (Nancy.Krebs@ucdenver.edu)
Dennis D Wallace (dwallace@rti.org)
Jonathan Swanson (jswanson5@gmail.com)
Marion Kos-Thomas (kosomari@mail.nih.gov)
Rexford Widmer (rexford.widmer@ge.com)
Robert L Goldenberg (rlg88@columbia.edu)

Version: 3  Date: 4 February 2014

Author’s response to reviews:

We have now addressed and included the names of the ethics committees which have reviewed and approved the trial in the study organization section.